Author's response to reviews

Title: Development and validation of sunlight exposure measurement questionnaire (SEM-Q) in an adult population residing in Pakistan.

Authors:

Quratulain Humayun (qh_khan@hotmail.com)
Romaina Iqbal (romaina.iqbal@aku.edu)
Iqbal Azam (iqbal.azam@aku.edu)
Amna Rehana Siddiqui (rehana.siddiqui@aku.edu)
Aysha Habib Khan (aysha.habib@aku.edu)
Naila Baig-ansari (naila@zazo.org)

Version: 3 Date: 15 May 2012

Author's response to reviews: see over
To
Dr George Rachiotis
Editor
Bio-Med Central

Re: MS 4487558236486080 Development and validation of sunlight exposure measurement questionnaire (SEM-Q) in an adult population residing in Pakistan.

Dear Dr. Rachiotis,
Thank you for sharing reviewer’s comments related to our manuscript on development and validation of a sunlight exposure measurement questionnaire with me. We have added a sentence, addressing the comment that the reviewer had made, in the manuscript on page 6, under the heading of development of scoring system for SEM-Q’s. I hope this addition of sentence will clarify things as we did multiply the scores in the manner the reviewer has described.

I hope you will find the revision satisfactory and accept our manuscript for publication in your esteemed journal. Please do not hesitate to contact me if you have any queries.

Thank you

Romaina Iqbal

Romaina Iqbal MSc, MPH, PhD
Assistant Professor
Departments of Community Health Sciences and Medicine
Aga Khan University
Phone Ext. 4832/4807